Author's response to reviews

Title: Shikonin suppresses ERK 1/2 phosphorylation during the early stages of adipocyte differentiation in 3T3-L1 cells

Authors:

Soyoung Gwon (herotigger@gmail.com)
Jiyun Ahn (jyan@kfri.re.kr)
Changhwa Jung (chjung@kfri.re.kr)
Bokyung Moon (bkmoon@cau.ac.kr)
Taeyoul Ha (tyhap@kfri.re.kr)

Version: 3 Date: 10 July 2013

Author's response to reviews: see over
July 10, 2013

Dear Editor

Thank you for giving us an opportunity to revise our manuscript. We revised the manuscript in accordance with your comments. As you suggested, our manuscript was edited by American Journal Experts (www.journalexperts.com) for grammar and spelling. Please find attached a revised version of our manuscript.

Sincerely yours,
Taeyoul Ha, Ph.D.
Principal researcher,
Functional Food Technology Research Group
Korea Food Research Institute
Tel.: 82-31-780-9054
Fax: 82-31-780-9225
E-mail: tyhap@kfri.re.kr